Author's response to reviews

Title: Case report: Scleroderma with crescentic glomerulonephritis

Authors:

Arunachalam Ramaswami (ramsdr@yahoo.com)
Thiraviam Kandaswamy (tknephro@yahoo.com)
Tholappan Rajendran (ammukkannu@yahoo.com)
kizhakkePisharam Jeyakrishnan (jkpish@yahoo.co.uk)
Hla Aung (hlaaung@brunet.bn)
Mohammad Iqbal (IQBAl103@hotmail.com)
Chakko K Jacob (ckjacob@hotmail.com)
Saukat Zinna (zinnashaukat@yahoo.com)
Gazala Kafeel (gbilgrami@hotmail.com)

Version: 2 Date: 16 July 2007

Author's response to reviews:

Dear Sirs, We are submitting a case report of rare presentation of a rare disease as it may have a direct bearing in the management. Please consider this for publication. Yours truly, Dr. A. Ramaswami

Peer reviewers: Prof. P. Soundera Rajan, HOD nephrology, Sri Ramachandra Institute of Medical Sciences, Chennai, India

Dr. Jeya Kumar, Prof of Nephrology Madras Medical College and Hospital, Chennai, India